ADIS DRUG EVALUATION



Efgartigimod: A Review in Generalised Myasthenia Gravis

Hannah A. Blair¹

Accepted: 17 September 2024 / Published online: 7 November 2024 © Springer Nature Switzerland AG 2024

Abstract

Efgartigimod (Vyvgart[®]; Vyvgart[®] Hytrulo) is a neonatal fragment crystallizable receptor (FcRn) antagonist indicated for the treatment of generalised myasthenia gravis (gMG) in adults who are acetylcholine receptor (AChR) antibody positive (Ab+). Efgartigimod is approved for both intravenous (IV) and subcutaneous (SC) use. In a pivotal phase III trial, IV efgartigimod was associated with significant and clinically meaningful improvements in myasthenia gravis symptoms and reductions in disease burden. The beneficial effects of IV efgartigimod were reproducible, durable and maintained over the long term. IV efgartigimod also improved health-related quality of life (HRQOL). In another phase III trial, SC efgartigimod PH20 was noninferior to IV efgartigimod in reducing total immunoglobulin G levels. Clinical improvement with SC efgartigimod PH20 was consistent with that of IV efgartigimod and was reproducible over the long term. Efgartigimod was generally well tolerated; the most common adverse events were headache and infections (with IV efgartigimod) and injection-site reactions (with SC efgartigimod PH20). Although further long-term data are required, IV and SC formulations of efgartigimod provide effective, generally well-tolerated and flexible treatment options for adults with AChR Ab+ gMG.

Plain Language Summary

Generalised myasthenia gravis (gMG) is a chronic, autoimmune disorder caused by impaired communication between the nerves and muscles. Recently, new targeted therapies have been developed for gMG, including FcRn antagonists such as efgartigimod (Vyvgart[®]; Vyvgart[®] Hytrulo). Efgartigimod works by reducing circulating levels of disease-causing antibodies. The drug is available as an IV infusion or an SC injection. In patients with gMG, IV efgartigimod significantly reduced disease burden and improved myasthenia gravis symptoms. These beneficial effects were long-lasting and repeatable across multiple treatment cycles. IV efgartigimod provided consistent clinically meaningful improvement over the long term (up to 17 treatment cycles) and was generally well tolerated. The efficacy and tolerability of SC efgartigimod PH20 was consistent with that of IV efgartigimod. The most common adverse events were headache and infections (with IV formulation) and injection-site reactions (with SC formulation). Thus, efgartigimod is an effective and generally well-tolerated treatment option for adults with gMG, with the flexibility of IV or SC administration.

Digital Features for this Adis Drug Evaluation can be found at https://doi.org/10.6084/m9.figshare.27048220.

The manuscript was reviewed by: *S. Jacob*, University Hospitals Birmingham NHS Foundation Trust, Queen Elizabeth Neuroscience Centre and Institute of Immunology and Immunotherapy, University of Birmingham, Birmingham, UK; *J. Morena*, Duke University Hospital, Durham, NC, USA; *S. M. Raja*, Department of Neurology, Duke University School of Medicine, Durham, NC, USA.

Hannah A. Blair demail@springer.com

1 Introduction

Myasthenia gravis is a chronic autoimmune disorder characterized by muscle weakness due to impaired neuromuscular transmission [1, 2]. It is rare, with an annual incidence of 3–28 cases per million. The disease can be broadly classified as ocular or generalised [1, 2]. The clinical defining feature of myasthenia gravis is fatigable muscle weakness, which may affect ocular, bulbar, limb and respiratory muscles [1–3]. The underlying pathogenesis of myasthenia gravis involves autoantibodies that bind to the acetylcholine receptor (AChR), muscle-specific tyrosine kinase (MuSK) or low-density lipoprotein receptor-related protein 4. In most (80–90%) cases of myasthenia gravis, immunoglobulin (Ig) G antibodies are directed against AChRs on the postsynaptic

¹ Springer Nature, Private Bag 65901, Mairangi Bay, Auckland 0754, New Zealand

H. A. Blair

Efgartigimod: clinical considerations in gMG

First FcRn antagonist to be approved for gMG

Available as IV and SC formulations

Improves myasthenia gravis symptoms, reduces disease burden and improves HRQOL in AChR Ab+ patients

Beneficial effects are reproducible, durable and maintained over the long term

Generally well tolerated

membrane of the neuromuscular junction [1-3]. These autoantibodies bind to AChRs, leading to complement-mediated destruction of AChRs and dysfunction of the neuromuscular junction [1, 2]. Pathogenic AChR autoantibodies also impair the function of AChR by preventing acetylcholine from binding to the receptor [2].

Traditionally, patients with myasthenia gravis have been treated with acetylcholinesterase inhibitors (AChEIs), corticosteroids, nonsteroidal immunosuppressive therapies (NSISTs), biologicals, intravenous immunoglobulins (IVIg), therapeutic plasma exchange (PLEX) and thymectomy [1–5]. However, 10–20% of patients do not respond to immunosuppressive therapy, which is also associated with treatment-limiting adverse events (AEs) [3, 4]. An improved understanding of the pathophysiology of myasthenia gravis has led to the development of new targeted therapeutic approaches [1–5]. One such approach involves inhibition of the neonatal fragment crystallizable receptor (FcRn), which maintains IgG levels by protecting antibodies from lysosomal degradation and recycling them back into the circulation [1–5].

Intravenous (IV) efgartigimod (Vyvgart[®]) was the first FcRn antagonist to be approved for the treatment of generalised myasthenia gravis (gMG) [6, 7]. A subcutaneous (SC) formulation of efgartigimod co-formulated with recombinant human hyaluronidase PH20 (Vyvgart® Hytrulo; hereafter referred to as SC efgartigimod PH20) is now also available. Both formulations of efgartigimod are approved in several regions worldwide, including the USA [8, 9], the EU [10] and China [11], for the treatment of gMG in adults who are AChR antibody positive (Ab+), and in Japan (VYVDURA [®]), for the treatment of gMG in adults with an inadequate response to treatment with steroids or NSISTs, regardless of AChR antibody status [12]. This article reviews the efficacy and tolerability of IV efgartigimod and SC efgartigimod PH20 in this patient population and briefly summarizes the pharmacological properties of efgartigimod.

2 Pharmacodynamics of Efgartigimod

Efgartigimod is a humanized IgG1-derived Fc fragment of the za allotype [8, 9, 13]. It has been engineered for increased affinity to FcRn at both neutral and acidic pH, with an equilibrium dissociation constant (K_D) of 320 nM at pH 7.4 and a K_D of 14.2 nM at pH 6.0 [13]. Binding of efgartigimod to FcRn inhibits its interaction with IgG, thereby reducing levels of circulating IgG (including pathogenic IgG autoantibodies) [10, 13]. This targeted reduction of IgG levels occurs without reducing levels of other immunoglobulins or albumin and without increasing cholesterol levels [10, 13].

IV efgartigimod rapidly reduced serum IgG levels in a murine model of MuSK myasthenia gravis [14] and in cynomolgus monkeys [13]. IV efgartigimod also improved muscle weakness and fatigability in myasthenic mice relative to control [14]. In phase I trials in healthy volunteers, single and multiple ascending doses of IV efgartigimod [13] and SC efgartigimod PH20 [15] reduced total IgG levels from baseline, but did not alter serum levels of other immunoglobulins or albumin [13]. Total IgG levels returned to baseline ≈ 8 weeks after the last IV infusion [13].

In an exploratory phase II trial [16] and in the phase III ADAPT trial [17] (Sect. 4.1.1) in AChR Ab+ patients with gMG, four once-weekly IV infusions of efgartigimod 10 mg/kg per cycle rapidly decreased serum IgG levels and AChR antibody levels from baseline. Rapid and sustained IgG reductions were observed across all subtypes of IgG and during subsequent treatment cycles [16, 17]. Total IgG and AChR antibody levels returned to baseline levels \approx 7–8 weeks after the last IV infusion [16, 17]. In the ADAPT+ open-label extension (OLE; Sect. 4.1.1.2), reductions in total IgG, IgG subtypes and AChR antibody levels were consistent with those seen in ADAPT [18].

In a phase I trial in healthy volunteers, SC efgartigimod PH20 had similar pharmacodynamic effects as IV efgartigimod in terms of reducing total IgG levels [15]. The phase III ADAPT-SC trial (Sect. 4.2.1) confirmed the pharmacodynamic noninferiority of SC efgartigimod PH20 1000 mg to IV efgartigimod 10 mg/kg in patients with gMG [19]. The primary endpoint of ADAPT-SC was the percentage change from baseline in total IgG levels at week 4 and the prespecified noninferiority margin was 10%. At week 4, after one cycle of treatment (four once-weekly administrations), the least squares mean (LSM) change from baseline in total IgG level was -66.4% in patients who received SC efgartigimod PH20 and -62.2% in patients who received IV efgartigimod (LSM difference -4.2% 95% CI -7.73 to -0.66; p < 0.0001 for noninferiority). Similar results were seen in the AChR Ab+ population (-66.9% vs -62.4%; LSM difference -4.5%; 95% CI -8.53 to -0.46; p < 0.0001). Total IgG

levels returned to near baseline levels by week 10. Reductions from baseline in AChR antibody levels were similar in the SC and IV treatment groups and paralleled the reductions in total IgG [19].

3 Pharmacokinetics of Efgartigimod

Efgartigimod exhibits linear pharmacokinetics [8, 10], with a geometric mean accumulation ratio of 1.12 [10]. Exposure of efgartigimod increases in a dose-proportional manner following IV doses up to 50 mg/kg (i.e. 5 x the recommended dosage) [8] and SC doses up to 1750 mg (i.e. 1.75 x the recommended dosage) [9]. In the ADAPT-SC trial in patients with gMG (Sect. 4.2.1), exposure of IV efgartigimod 10 mg/kg and SC efgartigimod PH20 1000 mg was comparable following one treatment cycle (four once-weekly administrations) [15]. The estimated bioavailability of SC efgartigimod PH20 is 77% [9]. The volume of distribution of efgartigimod is 15–20 L [8–10].

The expected metabolic pathway for efgartigimod (and hyaluronidase [9]) involves degradation to small peptides and amino acids by proteolytic enzymes [8–10]. Following a single dose of IV efgartigimod 10 mg/kg in healthy volunteers, <0.1% of the dose was recovered in urine [8, 9]. Efgartigimod has a terminal half-life of 80–120 h (3–5 days) [8–10] and a clearance of 0.128 L/h [10].

Age, sex, race and body weight did not have any clinically relevant effects on efgartigimod pharmacokinetics [8–10, 15]. Dedicated pharmacokinetic studies have not been conducted in patients with renal or hepatic impairment [8–10]. However, in a population pharmacokinetic analysis, patients with mild renal impairment [estimated glomerular filtration rate (eGFR) 60–89 mL/min/1.73 m²] had a 22% (IV efgartigimod) or 11% (SC efgartigimod PH20) increase in efgartigimod exposure relative to that in patients with normal renal function [8, 9]. The effect of moderate (eGFR 30–59 mL/min/1.73 m²) and severe (eGFR < 30 mL/min/1.73 m²) renal impairment on the pharmacokinetics of efgartigimod is unknown [10]. Hepatic impairment is not expected to influence efgartigimod pharmacokinetics [8–10].

3.1 Drug Interactions

No formal drug interaction studies have been conducted with efgartigimod [8–10]. However, because efgartigimod is not metabolized by CYP450 enzymes, interactions between efgartigimod and concomitant drugs that are substrates, inducers or inhibitors of CYP450 enzymes are not expected [8, 9]. Efgartigimod may decrease exposure to drugs that bind to the human FcRn (e.g. immunoglobulin products, monoclonal antibodies, antibody derivatives containing the

human FcRn domain of the IgG subclass). Plasma exchange, plasmapheresis and immunoadsorption may decrease circulating concentrations of efgartigimod [10].

Due to its mechanism of action (e.g. reduction of IgG levels), efgartigimod has the potential to interfere with the response to vaccines [8–10]. However, healthy volunteers who received IV efgartigimod mounted effective humoral immune responses to a pneumococcal vaccine [20]. Likewise, IV efgartigimod-treated patients with gMG participating in the ADAPT (Sect. 4.1.1) or ADAPT+ (Sect. 4.1.1.2) trials mounted antigen-specific IgG responses to pneumococcal, influenza and COVID-19 vaccines [21]. Administration of live or live-attenuated vaccines during efgartigimod treatment is not recommended [8–10].

4 Therapeutic Efficacy of Efgartigimod

4.1 Intravenous Efgartigimod

The efficacy of IV efgartigimod in patients with gMG was initially demonstrated in an exploratory, multinational, randomized, double-blind, phase II trial [16]. In this trial, relative to placebo, patients who received four once-weekly IV infusions of efgartigimod 10 mg/kg demonstrated rapid and significant clinical improvement (assessed on four efficacy scales). Most patients achieved persistent (≥ 6 weeks) disease improvement [16]. Based on these findings, the efficacy of efgartigimod was subsequently evaluated in the multinational, randomized, double-blind, phase III ADAPT trial [17] (Sect. 4.1.1). These data are supported by the long-term results of ADAPT+, an up to 3-year OLE of ADAPT [18] (Sect. 4.1.1.2). The effectiveness of IV efgartigimod in the real-world setting is also discussed (Sect. 4.1.2).

4.1.1 ADAPT Trial

The ADAPT trial enrolled patients aged ≥ 18 years with gMG, with or without AChR antibodies [17]. Inclusion criteria were Myasthenia Gravis Foundation of America (MGFA) class II-IV disease; a Myasthenia Gravis Activities of Daily Living (MG-ADL) score of ≥ 5 (> 50% non-ocular); and a history of abnormal neuromuscular transmission tests, a positive edrophonium chloride test, or improvement with AChEIs. All patients were receiving stable doses of ≥ 1 gMG treatment (i.e. NSISTs, steroids or AChEIs). Following stratification based on AChR antibody status (positive vs negative), concomitant NSISTs (yes vs no) and Japanese nationality (yes vs no), patients were randomized to receive a treatment cycle comprising four once-weekly IV infusions of efgartigimod 10 mg/kg (n = 84) or placebo (n = 83). After the first cycle, patients could receive one or two additional cycles according to individual clinical response (i.e.

Table 1 Efficacy of intravenous efgartigimod in patients with generalised myasthenia gravis in the phase III ADAPT trial					
ADAPT trial [17]	IV EFG 10 mg/kg ($n = 65$)	PL $(n = 64)$	OR (95% CI)		
Primary endpoint					
MG-ADL response ^a in cycle 1 (% of pts)	68*	30	4.95 (2.21–11.53)		
Secondary endpoints ^b					
QMG response ^c in cycle 1 (% of pts)	63*	14	10.84 (4.18-31.20)		
MG-ADL response ^d in cycle 1 in overall population (% of pts)	68*	37	3.70 (1.85-7.58)		
Proportion of time with CMI ^e up to day 126 (%)	48.7*	26.6			
Median time from day 28 until no CMI (days)	35	8			
Early MG-ADL ^f response in cycle 1 ^g (% of pts)	57	25			

Unless otherwise stated, all efficacy endpoints were assessed in the modified intention-to-treat population of AChR Ab+ pts

Ab+ antibody positive, Ab- antibody negative, AChR acetylcholine receptor, CMI clinically meaningful improvement, EFG efgartigimod, MG-ADL Myasthenia Gravis Activities of Daily Living, OR odds ratio, PL placebo, pts patients, QMG Quantitative Myasthenia Gravis *p < 0.0001 vs PL

 $^{a}A \ge 2$ -point improvement in MG-ADL score sustained for ≥ 4 consecutive weeks, with the first improvement occurring by week 4

^bSecondary endpoints were assessed hierarchically in the order presented

^cA \geq 3-point improvement in total QMG score for \geq 4 consecutive weeks, with the first improvement occurring by week 4

^dAll randomized pts, including AChR Ab– pts (EFG n = 84 and PL n = 83)

 $^{e}A \ge 2$ -point improvement from baseline in MG-ADL score

 $^{f}A \ge 2$ -point improvement in MG-ADL score sustained for ≥ 4 consecutive weeks, with the first improvement occurring by week 2

^gStatistical analysis was not performed due to the lack of statistical significance for the previous secondary endpoint

MG-ADL score of ≥ 5 and when an MG-ADL responder no longer had a ≥ 2 -point improvement from baseline in MG-ADL total score). Subsequent cycles could commence ≥ 8 weeks after initiation of the previous cycle [17].

The primary endpoint was the proportion of AChR Ab+ patients with an MG-ADL response in the first treatment cycle [17]. An MG-ADL response was defined as a \geq 2-point improvement (reduction) in MG-ADL score sustained for \geq 4 consecutive weeks, with the first improvement by week 4 (i.e. 1 week after the last infusion of the cycle). Baseline demographic and clinical characteristics were generally well balanced between treatment groups. However, more patients in the efgartigimod group had previously undergone thymectomy than in the placebo group (70% vs 43%). In these patients, the median time since thymectomy was 10.8 years. Most patients were female (71%), had MGFA class II or III disease (96%), were AChR Ab+ (77%) and were receiving immunosuppressive treatment (86%) [17].

IV efgartigimod was more effective than placebo in improving myasthenia gravis symptoms and reducing disease burden in AChR Ab+ patients with gMG [17]. Significantly more efgartigimod than placebo recipients achieved an MG-ADL response in the first cycle (primary endpoint; Table 1). Efgartigimod was also significantly more effective than placebo for the first three hierarchically tested secondary endpoints: the proportion of patients achieving a Quantitative Myasthenia Gravis (QMG) response in the first cycle; the proportion of MG-ADL responders in the overall patient population in the first cycle; and the proportion of time with clinically meaningful (i.e. \geq 2-point) improvement (CMI) in MG-ADL score up to day 126 (Table 1). The median time from day 28 until no CMI was not significantly different between treatment groups; therefore, statistical analysis of the next secondary endpoint (the proportion of patients with an early MG-ADL response in the first cycle) was not performed (Table 1). Clinical improvement was seen as early as 1 week after the first infusion [17].

Overall, more IV efgartigimod than placebo recipients achieved higher levels of improvement in MG-ADL score (up to 9-point reduction) and QMG score (up to 10-point reduction) by week 4 [17]. The proportion of patients achieving minimal symptom expression (MSE; i.e. MG-ADL score of 0 or 1) was significantly higher with efgartigimod than with placebo (40% vs 11%; p < 0.0001). Efgartigimod recipients had significantly (p < 0.05) greater improvements from baseline in MG-ADL, QMG and Myasthenia Gravis Composite (MGC) scores during the first cycle than placebo recipients; these improvements were seen as early as week 1 and were sustained through week 7 [17]. In a post hoc analysis, efgartigimod was associated with a significant reduction in the risk of exacerbation (i.e. a 3-point worsening in QMG score; 21% vs 44%; p = 0.0016) and numerically lower rates of all-cause hospitalization [11.4 vs 28.3 per 100 patientyears (PY)] and gMG-related hospitalization (2.8 vs 8.5 per 100 PY) compared with placebo [22].

The clinical benefit of IV efgartigimod was reproducible after subsequent cycles [17]. Among patients who received a second cycle, 71% of efgartigimod recipients achieved an MG-ADL response compared with 26% of placebo recipients. Of the 29 patients who did not respond to efgartigimod during cycle 1, 19 were retreated and seven (37%) achieved an MG-ADL response during cycle 2 [17].

According to predefined exploratory and post hoc subgroup analyses, the benefits of IV efgartigimod over placebo were seen regardless of sex [17, 23], age [17], baseline MG-ADL scores [17], concomitant medications (NSISTs, steroids and/or AChEIs) [17, 24], history of thymectomy [17], affected muscle group subdomains (bulbar, ocular, limb/gross motor, respiratory) [25], prior treatment failures [26], disease duration [27] and body weight [28]. However, among AChR Ab– patients, the proportion of MG-ADL responders was similar in both treatment groups (68% with efgartigimod and 63% with placebo) [17]. The proportion of patients achieving a QMG response was 53% with efgartigimod and 37% with placebo [17].

4.1.1.1 Health-Related Quality of Life IV efgartigimod was associated with rapid, significant, durable and reproducible improvements in health-related quality of life (HRQOL) in AChR Ab+ patients with gMG [29]. HROOL was assessed using the Myasthenia Gravis-Quality of Life 15-item revised (MG-QOL15r), a disease-specific measure assessing mobility, disease symptoms, general contentment and emotional well-being, with higher scores indicating worse quality of life, and the EuroQol 5-Dimensions 5-Levels (EQ-5D-5L), a standardized measure of health status, with higher scores indicating better quality of life. Efgartigimod was associated with significantly greater reduction in MG-QOL15r scores and significantly greater increase in EQ-5D-5L utility scores compared with placebo. Significant betweengroup differences were seen as early as week 1 (p < 0.01) and were maintained for up to 8 weeks (p < 0.05) in both cycle 1 and cycle 2. EQ-5D-5L visual analogue scale (VAS) scores were also significantly increased with efgartigimod relative to placebo, with these improvements maintained up to week 5 in both treatment cycles [29]. Improvements in MG-ADL scores with efgartigimod were associated with higher EQ-5D-5L utility scores, with each unit improvement in MG-ADL leading to a significant (p < 0.001) utility increase of 0.0233 [30].

4.1.1.2 Long-Term Efficacy The beneficial effects of IV efgartigimod in patients with gMG were maintained over the long term [18]. Patients who completed the ADAPT trial (or who were eligible for a treatment cycle but could not complete the cycle by week 26 of ADAPT) were eligible to enter ADAPT+. The primary and key secondary objectives were to evaluate long-term safety and tolerability (Sect. 5);

efficacy was also assessed. All patients received four onceweekly IV infusions of efgartigimod 10 mg/kg per cycle. The number of treatment cycles was determined according to individual clinical response [18]. Patients who completed ADAPT+ were eligible to enter ADAPT-SC+ (Sect. 4.2.2).

At the time of the final interim analysis (data cutoff 31 January 2022), 145 patients had received ≥ 1 dose of efgartigimod [18]. The mean duration of treatment plus follow-up was 548 days and patients received a maximum of 17 treatment cycles, corresponding to 217.6 PY of observation. In AChR Ab+ patients, the mean change in MG-ADL and QMG scores during the first cycle was -5.0 and -4.7, respectively. Similar results were seen in AChR Ab+ patients (-5.3 and -5.2, respectively). Most AChR Ab+ patients had CMI in MG-ADL (i.e. ≥ 2 -point reduction) and QMG (i.e. ≥ 3 -point reduction) that were maintained in each cycle for up to 10 cycles, with a substantial proportion of patients demonstrating improvements well beyond clinically meaningful thresholds [18].

In a post hoc analysis in AChR Ab+ patients with ≥ 1 year of combined follow-up between ADAPT and ADAPT+ (n = 95), sustained clinical improvement with IV efgartigimod was associated with fewer treatment cycles [18]. In the entire cohort, the annualized mean number of cycles was 4.7 cycles per year. The average time between cycles (i.e. from the last infusion of the previous cycle to the first infusion of the subsequent cycle) was ≥ 9 weeks in 37% of patients (≈ 4 cycles per year). Overall, 24% of patients received ≤ 3 cycles per year, 18% received ≤ 2 cycles per year and 6% received 1 cycle per year [18].

4.1.2 In the Real-World Setting

Data from the real-world setting support the efficacy of IV efgartigimod for the treatment of gMG. Clinical experience in the USA [31–33], the UK [34], Italy [35, 36], China [37] and Japan [38] confirmed that weekly infusions of IV efgartigimod led to CMI in patients with gMG.

In the largest of these studies, which was conducted in the USA, data were obtained from 705 patients with gMG who initiated IV efgartigimod and enrolled in a patient support programme [32]. At baseline, all patients had an MG-ADL score of ≥ 2 (mean 8.6). The number of efgartigimod cycles initiated ranged from 1 to 13 (mean 3.8). The largest observed 'cohort-wide response' was a 5.8-point mean reduction from baseline in MG-ADL score with 93% of patients experiencing CMI. The largest observed MG-ADL responses were 78.8%, 67.4%, 61.0% and 57.1% in the bulbar, ocular, limb/gross motor and respiratory subdomains, respectively. At their best state, 78% of patients had achieved an MG-ADL score of ≤ 4 and 36% of patients had achieved MSE [32]. In a retrospective cohort study of US insurance claimsbased data in patients with gMG (n = 316), IV efgartigimod was associated with a significant reduction in oral corticosteroid usage [39]. Six months after initiating efgartigimod, the average oral corticosteroid dosage had significantly (p < 0.001) decreased from 18.6 mg/day to 13.5 mg/day. Almost half (46%) of patients were able to reduce their average oral corticosteroid dosage by ≥ 5 mg/day [39].

4.2 Subcutaneous Efgartigimod PH20

The efficacy of SC efgartigimod PH20 in patients with gMG was investigated in the randomized, open-label, multicentre, phase III ADAPT-SC trial [19] (Sect. 4.2.1), which was primarily designed to evaluate the pharmacodynamic noninferiority of SC efgartigimod PH20 compared with IV efgartigimod (Sect. 2). These data are supported by the interim results of ADAPT-SC+, an ongoing OLE of ADAPT-SC, which was designed to assess the long-term tolerability of SC efgartigimod PH20 [19, 40] (Sect. 4.2.2).

4.2.1 ADAPT-SC Trial

ADAPT-SC enrolled patients aged ≥ 18 years with MGFA class II–IV gMG and an MG-ADL total score of ≥ 5 (> 50% non-ocular) who were receiving a stable dose of ≥ 1 gMG treatment (NSISTs, steroids or AChEIs) [19]. They were randomized to receive SC efgartigimod 1000 mg co-formulated with PH20 2000 U/mL (n = 55) or IV infusions of efgartigimod 10 mg/kg (n = 55) once weekly for 4 weeks (i.e. one treatment cycle) followed by 7 weeks of post-treatment follow-up. Most (83%) patients were AChR Ab+ [19].

Clinical improvement was consistent between SC efgartigimod PH20 and IV efgartigimod in patients with gMG [19]. In the overall population, the proportions of MG-ADL and QMG responders were similar with SC efgartigimod PH20 versus IV efgartigimod (Table 2). The mean change from baseline in MG-ADL and QMG total scores at week 4 was similar in the SC and IV treatment groups (Table 2). The proportion of patients achieving MSE at any time point was also similar in both treatment groups. CMI was seen as early as week 1 in both treatment groups, with maximal improvement in MG-ADL and QMG scores observed at week 4. Up to one-third of patients in both treatment groups experienced MG-ADL and QMG improvements well beyond clinically meaningful thresholds (Table 2) [19].

Results in the AChR Ab+ population (n = 91) were similar to those in the overall population in terms of MG-ADL and QMG response rates, change from baseline in MG-ADL and QMG total scores and achievement of MSE and CMI (Table 2) [19]. Modest clinical improvements were also seen in AChR Ab- patients (n = 19). In this population, the MG-ADL response rate was 60% with SC efgartigimod PH20 and

56% with IV efgartigimod; corresponding QMG response rates were 50% and 44% [19].

4.2.2 Long-Term Efficacy

Long-term (up to 3 years) treatment with SC efgartigimod PH20 was associated with early, consistent and reproducible clinical improvement in patients with gMG, according to interim results of ADAPT-SC+ [19]. Patients who completed ADAPT-SC (n = 105) or ADAPT+ (n = 73) were eligible to enter ADAPT-SC+. All patients received individualized treatment cycles of four once-weekly administrations of SC efgartigimod 1000 mg co-formulated with PH20 2000 U/mL. Patients could start a new treatment cycle without a worsening MG-ADL score [19].

Through January 2022, a total of 164 patients received ≥ 1 dose of SC efgartigimod PH20, with a mean duration of follow-up of 169.7 days (72.1 PY of exposure) [19]. In the overall population, the mean change from baseline in MG-ADL total score at week 4 was -4.0, -3.8 and -4.1 in cycles 1 through 3, respectively. The proportions of patients achieving MSE at any time point were 30%, 31% and 36% in cycles 1 through 3, respectively. Most (88%) patients in ADAPT-SC+ were considered to be adequately trained in self-administration of SC efgartigimod PH20, with nearly 60% of patients able to self-administer at home by cycle 3. Among patients who had previously received IV efgartigimod in ADAPT+ or ADAPT-SC, 71% preferred SC efgartigimod PH20 and 29% preferred IV efgartigimod or had no preference [19].

The long-term efficacy of SC efgartigimod PH20 was also demonstrated in the AChR Ab+ (n = 141) [40] and AChR Ab- (n = 38) populations [41], at a later data cutoff of December 2022 (mean study duration of 412.9 days; maximum 585 days) [40]. Across cycles 1–9, the proportion of AChR Ab+ patients achieving MSE ranged from 36% to 44%, the proportion achieving CMI in MG-ADL at week 4 ranged from 76% to 82% and the mean change from baseline in MG-ADL score at week 4 ranged from -4.1 to -4.7 [40]. Among AChR Ab- patients, 68–90% experienced CMI in MG-ADL and 12–19% achieved MSE across multiple cycles [41].

SC efgartigimod PH20 was associated with consistent and reproducible improvements in HRQOL (assessed using the MG-QOL15r and EQ-5D-5L VAS) in both the AChR Ab+ [42] and AChR Ab- [41] populations.

5 Tolerability of Efgartigimod

IV efgartigimod was generally well tolerated in patients with gMG [17, 18]. In ADAPT, the incidence of AEs was 77% with IV efgartigimod and 84% with placebo [17].

1	469

Table 2 Efficacy of subcutaneous efgartigimod in patients with generalised myasthenia gravis in the phase III ADAPT-SC trial						
ADAPT-SC trial [19]	Overall population		AChR Ab+ population			
	SC EFG PH20 (<i>n</i> = 55)	IV EFG (<i>n</i> = 55)	SC EFG PH20 (<i>n</i> = 45)	IV EFG (<i>n</i> = 46)		
MG-ADL response ^a (% of pts)	69	69	71	72		
QMG response ^b (% of pts)	66	52	69	53		
Mean change from BL in MG-ADL total score at week 4	-5.1	-4.7	-5.3	-4.6		
Mean change from BL in QMG total score at week 4	-6.1	-5.2	-6.5	-5.4		
MSE ^c at any time point (% of pts)	37	38	46	41		
CMI in MG-ADL ^d at week 4 (% of pts)	90	91	93	91		
CMI in QMG ^e at week 4 (% of pts)	81	71	84	81		
CMI in MG-ADLd at week 10 (% of pts)	57	63	62	67		
\geq 7-point improvement in MG-ADL total score at any time point (% of pts)	25	30				
\geq 8-point improvement in QMG total score at any time point (% of pts)	33	29				

Efficacy analyses were performed in the intention-to-treat population

Ab+ antibody positive, AChR acetylcholine receptor, BL baseline, CMI clinically meaningful improvement, EFG efgartigimod, IV intravenous, MG-ADL Myasthenia Gravis Activities of Daily Living, MSE minimal symptom expression, pts patients, QMG Quantitative Myasthenia Gravis ^aA \geq 2-point reduction from BL in MG-ADL score for \geq 4 consecutive weeks after onset, with onset occurring \geq 1 week after last dose

^bA \geq 3-point reduction from BL in QMG score for \geq 4 consecutive weeks after onset, with onset occurring \geq 1 week after last dose ^cMG-ADL total score of 0 or 1

 $^{d}A \ge 2$ -point reduction from BL in MG-ADL total score

 $^{e}A \ge 3$ -point reduction from BL in QMG score

Most AEs were mild to moderate in severity. The most common (incidence $\geq 10\%$ of patients and higher than with placebo) AEs with efgartigimod were headache (29% vs 28% with placebo), upper respiratory tract infection (11% vs 5%) and urinary tract infection (10% vs 5%). Serious AEs occurred in 5% of efgartigimod recipients and 8% of placebo recipients, while AEs leading to discontinuation of treatment occurred in 4% of patients in both groups [17].

SC efgartigimod PH20 and IV efgartigimod were generally well tolerated in ADAPT-SC [19]. The overall incidence of AEs was 67% with SC efgartigimod PH20 and 51% with IV efgartigimod. Most AEs were mild to moderate in severity. The most common (incidence $\geq 10\%$ of patients) AEs were injection-site reactions (ISRs; 38% with SC efgartigimod PH20 vs 2% with IV efgartigimod), headache (13% vs 13%), COVID-19 (4% vs 0%) and myasthenia gravis (11% vs 2%). The incidence of serious AEs was 15% with SC efgartigimod PH20 and 7% with IV efgartigimod, and 4% of patients discontinued SC efgartigimod PH20 due to AEs [19].

5.1 Long-Term Tolerability

The long-term tolerability profile of IV efgartigimod was generally consistent with that seen in short-term trials [18]. Over 217.5 PY of follow-up (data cutoff 31 January 2022) in ADAPT+, the incidence of treatment-emergent AEs (TEAEs) was 85% (3.60 events per PY of follow-up).

The most common (incidence $\geq 10\%$ of patients) TEAEs were infections (55%; 0.75 events/PY), headache (25%; 0.45 events/PY), COVID-19 (15%; 0.11 events/PY), nasopharyngitis (14%; 0.11 events/PY), infusion-related reactions (IRRs; 10%; 0.10 events/PY) and diarrhoea (10%; 0.09 events/PY). Serious TEAEs occurred in 23% of patients (0.24 events/PY) and 8% of patients discontinued treatment due to TEAEs (0.06 events/PY). There were five fatal TEAEs (0.02 events/PY), none of which were related to IV efgartigimod [18].

The long-term tolerability profile of SC efgartigimod PH20 was consistent with that of IV efgartigimod, with no new safety signals observed [40]. Over 193.4 PY of follow-up (data cutoff December 2022) in ADAPT-SC+, the incidence of AEs with SC efgartigimod PH20 was 85% (9.0 events/ PY). The most common (incidence $\geq 10\%$ of patients) AEs were injection-site erythema (29%; 1.7 events/PY), COVID-19 (22%; 0.2 events/PY), headache (20%; 0.6 events/PY), nasopharyngitis (16%; 0.2 events/PY), diarrhoea (13%; 0.2 events/PY), injection-site pain (12%; 0.2 events/PY), injection-site pruritus (11%; 0.2 events/PY) and injection-site bruising (10%; 0.2 events/PY). Grade ≥ 3 AEs occurred in 20% of patients (0.4 events/PY) and serious AEs occurred in 18% of patients (0.3 events/PY). Four patients discontinued treatment due to AEs. There were four fatal AEs (<0.1 events/PY), none of which were related to SC efgartigimod PH20 [40].

5.2 Adverse Events of Special Interest

Efgartigimod can increase the risk of infections [8–10], as can some concomitant medications (e.g. immunosuppressants) [43]. In ADAPT, infections (mostly mild to moderate in severity) occurred in 46% of IV efgartigimod recipients and 37% of placebo recipients [17]. In ADAPT+, the overall incidence of infections with IV efgartigimod was 55% (0.75 events/PY), and the incidence did not increase with subsequent treatment cycles [18]. In ADAPT-SC, infections (mostly mild to moderate in severity) occurred in 18% of patients in the SC efgartigimod PH20 group and 16% of patients in the IV efgartigimod group [19]. In ADAPT-SC+, the overall incidence of infections with SC efgartigimod PH20 was 51% (1.0 events/PY) [40]. Patients should be monitored for signs and symptoms of infection during treatment with efgartigimod [8-10]. In patients with an active infection, or if serious infection occurs, efgartigimod should be delayed or withheld until the infection has resolved [8-10].

IRRs, ISRs and hypersensitivity reactions may occur with efgartigimod [8-10]. In ADAPT, IRRs occurred in 4% of IV efgartigimod recipients and 10% of placebo recipients, all of which were mild in severity [17]. In ADAPT+, the incidence of IRRs with IV efgartigimod was 10% (0.10 events/PY) [18]. In ADAPT-SC, ISRs were the most commonly reported AE in the SC efgartigimod PH20 group (Sect. 5) [19]. All ISRs were mild or moderate in severity, and most were transient, resolved without treatment and did not lead to discontinuation of efgartigimod [19]. In ADAPT-SC+, the overall incidence of ISRs with SC efgartigimod PH20 was 46% (3.2 events/PY) [40]. There were no serious or grade \geq 3 ISRs and none led to discontinuation of treatment [40]. The incidence of ISRs decreased with each injection of the treatment cycle in ADAPT-SC (from 22% of patients with injection 1 to 10% with injection 4) [19] and over subsequent treatment cycles in ADAPT-SC+ (from 35% of patients in cycle 1 to 10% in cycle 9). Patients should be monitored for signs and symptoms of IRRs, ISRs and hypersensitivity reactions during treatment with efgartigimod [10]. Reactions should be treated with appropriate supportive therapies [8–10]. Subsequent injections/infusions should be administered with caution, with consideration given to close observation, slower infusion rates and/or pre-medications [8, 9].

Like all therapeutic proteins, efgartigimod has the potential for immunogenicity [8]. In ADAPT-SC, anti-drug antibodies (ADAs) were detected in 20% of patients following treatment with IV efgartigimod and in 35% of patients following treatment with SC efgartigimod PH20, with 4% of patients in both groups developing neutralizing antibodies [19]. In ADAPT+, 16% of patients developed ADAs following long-term treatment with IV efgartigimod, mostly during the first treatment cycle [18]. Although data are limited [8, 9], the presence of ADAs did not appear to affect the pharmacokinetics, efficacy or safety of efgartigimod [18, 19].

6 Dosage and Administration

In the USA, efgartigimod is indicated for the treatment of gMG in adult patients who are AChR Ab+ [8, 9]. In the EU [10] and in China [11], efgartigimod is indicated as an add-on to standard therapy in adult patients with gMG who are AChR Ab+. Efgartigimod is also approved in Japan for the treatment of gMG in adults with an inadequate response to treatment with steroids or NSISTs, regardless of AChR antibody status [12].

Efgartigimod is approved for IV or SC use [8–10]. IV efgartigimod is administered as a 1-h infusion once weekly for 4 weeks as one cycle; the recommended dosage is 10 mg/kg (or 1200 mg in patients weighing \geq 120 kg) [8, 10]. SC efgartigimod PH20 is injected over 30–90 s at a dosage of 1000 mg (1008 mg in the EU [10]) once weekly for 4 weeks as one cycle [9, 10]. Subsequent treatment cycles can be administered according to clinical evaluation [8–10]. The US prescribing information states that SC efgartigimod PH20 must be administered by a healthcare professional [9], while the EU summary of product characteristics states that patients or caregivers may administer SC efgartigimod PH20 at home after adequate training from a healthcare professional [10].

Dosage adjustments are not required in elderly patients, patients with mild renal impairment or patients with hepatic impairment [8–10]. Due to a lack of data, the use of efgartigimod during pregnancy or breastfeeding should only be considered if the clinical benefit outweighs the risks [8–10]. Consult local prescribing information for further details regarding contraindications, warnings and precautions, drug interactions and use in special populations.

7 Place of Efgartigimod in the Management of Generalised Myasthenia Gravis

The primary goal for the treatment of gMG, as appointed by the MGFA task force, is achievement of a minimal manifestation status (i.e. no symptoms or functional limitations but mild weakness in some muscles), with only mild AEs that do not require intervention [44]. International consensus guidelines for the management of myasthenia gravis recommend the AChEI pyridostigmine as first-line therapy, with corticosteroids reserved for patients who do not respond to an adequate trial of pyridostigmine. Other recommended subsequent-line treatments include NSISTs, IVIg and PLEX [44].

Novel targeted therapeutics with improved efficacy and favourable tolerability profiles may represent a new approach

to the management of gMG [4]. Along with efgartigimod [6, 7], the FcRn antagonist rozanolixizumab [45] and several complement inhibitors (i.e. eculizumab [46], ravulizumab [47] and zilucoplan [48]) have been approved for the treatment of gMG, and are generally used as second- or third-line options for patients with severe and/or refractory disease [49]. German [50] and Nordic [51] guidelines recommend efgartigimod as an option for the symptomatic second-line treatment of severe, refractory myasthenia gravis. As the most recent international treatment guidelines [44] were published prior to the approval of efgartigimod, updates are awaited with interest.

Approval of IV efgartigimod was based on data from the pivotal phase III ADAPT trial (Sect. 4.1.1). In this trial, IV efgartigimod was associated with significant and clinically meaningful improvements in myasthenia gravis symptoms and reductions in disease burden in patients with AChR Ab+ gMG. These benefits were observed early and were seen across multiple gMG-specific rating scales. Of note, over two-thirds of patients in the efgartigimod group achieved improvement beyond the clinically meaningful threshold (Sect. 4.1.1). The efficacy of IV efgartigimod was reproducible and durable, with consistent improvements observed with each subsequent treatment cycle (Sects. 4.1.1 and 4.1.1.2). Sustained duration of clinical improvement warranted fewer treatment cycles per year (Sect. 4.1.1.2), providing further support for the use of an individualized treatment approach [17, 18].

gMG can have a substantial negative impact on mental, physical and relational HRQOL [29, 30]. Patients with gMG often have comorbid depression and/or anxiety, both of which are predictors of worsening HRQOL [29]. Indeed, a substantial proportion of patients in the ADAPT trial reported anxiety/depression at baseline. Despite patients being on stable doses of ≥ 1 gMG treatment, baseline EQ-5D-5L health utility scores were also relatively low, highlighting the burden of disease [29]. IV efgartigimod led to rapid and significant improvements in HRQOL which were durable, reproducible and strongly correlated with improvements in clinical symptoms (Sect. 4.1.1.1).

Data from real-world studies of IV efgartigimod in patients with gMG were generally consistent with those seen in clinical trials (Sect. 4.1.2). Although treatment with IV efgartigimod was associated with CMI in the real-world setting, patient numbers in most studies were relatively small. It should also be noted that patients in the largest real-world study had an MG-ADL score of ≥ 2 at baseline (Sect. 4.1.2), while patients in the ADAPT trial had an MG-ADL score of ≥ 5 at baseline (Sect. 4.1.1). Patients with lower MG-ADL scores would be unlikely to attain MSE. Larger-scale and longer-term data are necessary to assess the effectiveness of efgartigimod in real-world clinical practice.

Conventional treatments for gMG are frequently associated with burdensome AEs that limit their use [3, 4]. Among the newer targeted treatments, complement inhibitors carry boxed warnings on the risk of life-threatening and fatal meningococcal infections [46–48]. IV efgartigimod was generally well tolerated in the ADAPT trial, with most AEs being mild or moderate in severity (Sect. 5). IV efgartigimod was also generally well tolerated over the long term (Sect. 5.1). The most common TEAEs in ADAPT+ were headache and infections (Sect. 5.1). Severe infections were uncommon, and the incidence of infection did not increase with subsequent cycles of IV efgartigimod (Sect. 5.2), possibly as a result of selective IgG reduction [18]. This is an important finding, given that patients with gMG are at increased risk of infections due to predisposition and the use of concomitant immunosuppressive therapy [17].

The US and EU labels for efgartigimod were extended to include SC efgartigimod PH20 [9, 10] based on data from the phase III ADAPT-SC trial (Sects. 2 and 4.2.1). In this trial, SC efgartigimod PH20 was noninferior to IV efgartigimod in reducing total IgG levels (Sect. 2) in patients with gMG, and clinical improvement was consistent between both formulations (Sect. 4.2.1). Interim results from the ongoing ADAPT-SC+ trial (Sect. 4.2.2) demonstrated that clinical improvement with SC efgartigimod PH20 was consistent and reproducible over the long term (≤ 3 years). The tolerability of SC efgartigimod PH20 was generally consistent with that of IV efgartigimod (Sect. 5). Almost half of patients experienced localized ISRs with the SC formulation (Sect. 5.2). However, most reactions were not serious or severe, did not result in treatment discontinuation and became less frequent over subsequent treatment cycles (Sect. 5.2).

More convenient administration has become the focus of recent development of gMG treatments [49]. Of the approved complement inhibitors, eculizumab and ravulizumab are both administered via IV infusion [46, 47], while zilucoplan can be self-administered once daily via SC injection [48]. Rozanolixizumab is administered as a SC infusion once weekly for 6 weeks [45] and other FcRn antagonists currently in development will likely require IV infusions every 2-4 weeks or SC injections every 1-2 weeks. Efgartigimod is approved for both IV and SC use (Sect. 6). The SC formulation was developed to provide patients with an alternative route of administration, allowing for a more individualized and flexible approach to treatment. SC efgartigimod PH20 may improve patient convenience, with delivery taking 30-90 s compared with a 1-h IV infusion (Sect. 6). More than two-thirds of patients in ADAPT-SC+ indicated a preference for SC efgartigimod PH20 over IV efgartigimod (Sect. 4.2.2). The SC formulation also offers the potential (after adequate training) for self- and caregiver-supported administration at home (Sect. 6). Indeed, by treatment cycle 3 of ADAPT-SC+, almost 60% of patients were able to self-administer SC efgartigimod PH20 at home

(Sect. 4.2.2). To further inform alternative treatment regimens, a randomized, open-label, phase IIIb trial (ADAPT-NXT) is assessing the efficacy and tolerability of continuous (10 mg/kg every 2 weeks) versus fixed cycles (four once-weekly infusions of 10 mg/kg per cycle) of IV efgartigimod in patients with gMG [52]. Final results are awaited with interest.

To date, no studies have directly compared efgartigimod with other gMG treatments. Systematic reviews, network meta-analyses and/or matching-adjusted indirect comparisons have demonstrated some differences in efficacy and tolerability between efgartigimod and other targeted therapies for gMG, including rozanolixizumab, complement inhibitors and B-cell therapies (e.g. belimumab, rituximab) [53–57]. However, given the limitations of indirect comparisons, these results should be interpreted with caution. Randomized head-to-head trials comparing the efficacy and tolerability of efgartigimod with other approved treatments would be valuable in determining their relative roles in the management of gMG.

The economic burden of gMG on both patients and caregivers is large [58]. A pharmacoeconomic model combining data from ADAPT and a real-world study found that add-on therapy with efgartigimod reduced total productivity losses per gMG patient by $\approx 27\%$ compared with conventional therapy alone [58]. In another analysis using data from phase III trials, the costs associated with achieving a 1-point improvement in QMG score, one additional patient with minimal clinically important difference in QMG, and one additional patient with MSE were lower for efgartigimod than for IVIg and eculizumab [59]. A Canadian cost-effectiveness analysis found that, over a lifetime horizon, IV efgartigimod was cost effective compared with chronic immunoglobulins [60]. However, according to the Institute for Clinical and Economic Review, efgartigimod has an incremental cost-effectiveness ratio that greatly exceeds the price needed to reach traditional cost-effectiveness thresholds [61]. The UK National Institute for Health Care and Excellence (NICE) does not recommend efgartigimod as a cost-effective use of National Health Service resources; however, the NICE guidance for efgartigimod in gMG is not yet finalized [62]. Given the high costs of FcRn antagonists (and complement inhibitors) [51, 61], further robust cost effectiveness analyses are warranted. A direct cost comparison of efgartigimod versus IVIg relative to symptom improvement would be of particular interest.

In conclusion, although further long-term data are required to fully determine the place of efgartigimod in the management of gMG, IV and SC formulations of efgartigimod provide effective, generally well-tolerated and flexible treatment options for adult patients with AChR Ab+ gMG.

Data Selection for Efgartigimod: 164 records identified

Duplicates removed	0
Excluded during initial screening (e.g. press releases; news reports; not relevant drug/indication; preclinical study; reviews; case reports; not randomized trial)	54
Excluded during writing (e.g. reviews; duplicate data; small patient number; nonrandomized/phase I/II trials)	48
Cited efficacy/tolerability articles	27
Cited articles not efficacy/tolerability	35
Search Strategy: EMBASE, MEDLINE and PubMed from 1	946

to present. Clinical trial registries/databases and websites were also searched for relevant data. Key words were efgartigimod, ARGX 113, ARGX-113, EFG PH20 SC, PH20 SC, Vyvgart, Hytrulo; generalized myasthenia gravis, myasthenia gravis, myasthenia, gMG, generalized MG. Records were limited to those in English language. Searches last updated 16 September 2024

Supplementary Information The online version contains supplementary material available at https://doi.org/10.1007/s40265-024-02101-9.

Acknowledgments During the peer review process, the manufacturer of efgartigimod was also offered an opportunity to review this article. Changes resulting from comments received were made on the basis of scientific and editorial merit.

Declarations

Funding The preparation of this review was not supported by any external funding.

Authorship and Conflict of Interest Hannah Blair is a salaried employee of Adis International Ltd/Springer Nature, and declares no relevant conflicts of interest. All authors contributed to this article and are responsible for its content.

Ethics Approval, Consent to Participate, Consent to Publish, Availability of Data and Material, Code Availability Not applicable.

References

- 1. Crisafulli S, Boccanegra B, Carollo M, et al. Myasthenia gravis treatment: from old drugs to innovative therapies with a glimpse into the future. CNS Drugs. 2024;38(1):15–32.
- Menon D, Bril V. Pharmacotherapy of generalized myasthenia gravis with special emphasis on newer biologicals. Drugs. 2022;82(8):865–87.
- Vanoli F, Mantegazza R. Antibody therapies in autoimmune neuromuscular junction disorders: approach to myasthenic crisis and chronic management. Neurotherapeutics. 2022;19(3):897–910.
- Alhaidar MK, Abumurad S, Soliven B, et al. Current treatment of myasthenia gravis. J Clin Med. 2022;11(6):1597.

- 5. Lascano AM, Lalive PH. Update in immunosuppressive therapy of myasthenia gravis. Autoimmun Rev. 2021;20(1): 102712.
- 6. Heo Y-A. Efgartigimod: first approval. Drugs. 2022;82(3):341–8.
- Heo Y-A. Efgartigimod alfa in generalised myasthenia gravis: a profile of its use. CNS Drugs. 2023;37(5):467–73.
- argenx US. VYVGART[®] (efgartigimod alfa-fcab) injection, for intravenous use: US prescribing information. 2023. https://daily med.nlm.nih.gov/dailymed/drugInfo.cfm?setid=8aefc8e3-26d6-4ff6-aab9-a7542927e084. Accessed 18 Sep 2024.
- argenx US. VYVGART[®] HYTRULO (efgartigimod alfa and hyaluronidase-qvfc) injection, for subcutaneous use: US prescribing information. 2024. https://dailymed.nlm.nih.gov/dailymed/ drugInfo.cfm?setid=7104d9e5-9910-47f5-9a93-741ac50f43b6. Accessed 18 Sep 2024.
- argenx. Vyvgart (efgartigimod alfa): EU summary of product characteristics. 2024. https://www.ema.europa.eu/en/medicines/ human/EPAR/vyvgart. Accessed 18 Sep 2024.
- 11. Pharmaceutical Technology. China NMPA approves Zai Labargenx's efgartigimod alfa for gMG [media release]. 17 Jul 2024. https://www.pharmaceutical-technology.com.
- Pharmaceutical Technology. argenx's VYVDURA gains approval in Japan for gMG treatment [media release]. 19 Jan 2024. https:// www.pharmaceutical-technology.com.
- Ulrichts P, Guglietta A, Dreier T, et al. Neonatal Fc receptor antagonist efgartigimod safely and sustainably reduces IgGs in humans. J Clin Invest. 2018;128(10):4372–86.
- Huijbers MG, Plomp JJ, van Es IE, et al. Efgartigimod improves muscle weakness in a mouse model for muscle-specific kinase myasthenia gravis. Exp Neurol. 2019;317:133–43.
- Vu T, Howard Jr J, Li Y, et al. Dose selection and clinical development of efgartigimod PH20 subcutaneous in patients with generalized myasthenia gravis [poster no. 270]. In: American Association of Neuromuscular & Electrodiagnostic Medicine (AANEM) Annual Meeting. 2023.
- Howard JF Jr, Bril V, Burns TM, et al. Randomized phase 2 study of FcRn antagonist efgartigimod in generalized myasthenia gravis. Neurology. 2019;92(23):e2661–73.
- Howard JF Jr, Bril V, Vu T, et al. Safety, efficacy, and tolerability of efgartigimod in patients with generalised myasthenia gravis (ADAPT): a multicentre, randomised, placebo-controlled, phase 3 trial. Lancet Neurol. 2021;20(7):526–36.
- Howard JF Jr, Bril V, Vu T, et al. Long-term safety, tolerability, and efficacy of efgartigimod (ADAPT+): interim results from a phase 3 open-label extension study in participants with generalized myasthenia gravis. Front Neurol. 2023;14(1284444):1–14.
- Howard JF Jr, Vu T, Li G, et al. Subcutaneous efgartigimod PH20 in generalized myasthenia gravis: a phase 3 randomized noninferiority study (ADAPT-SC) and interim analyses of a long-term open-label extension study (ADAPT-SC+). Neurotherapeutics. 2024. https://doi.org/10.1016/j.neurot.2024.e00378.
- 20. Azar A, Sleasman JW, Verhamme F, et al. Humoral immune response to polyvalent pneumococcal vaccine in healthy participants receiving efgartigimod [poster no. 1]. In: American Association of Neuromuscular & Electrodiagnostic Medicine (AANEM) Annual Meeting. 2023.
- Guptill JT, Sleasman JW, Steeland S, et al. Effect of FcRn antagonism on protective antibodies and to vaccines in IgG-mediated autoimmune diseases pemphigus and generalised myasthenia gravis. Autoimmunity. 2022;55(8):620–31.
- 22. Qi C, Dewilde S, Gelinas D, et al. Hospitalization and exacerbation estimates of efgartigimod vs. conventional therapy in generalized myasthenia gravis patients: a post-hoc analysis of the phase 3 ADAPT study [abstract no. CO135 plus poster]. Value Health. 2022;25(7):S329.
- 23. Hoffmann S, Zhao S, Callewaert E, et al. Gender-specific analysis of efgartigimod efficacy in patients with gMG: subanalysis

of the randomised phase 3 ADAPT trial [abstract no. EPO-055 plus poster]. Eur J Neurol. 2023;30(Suppl 1):358.

- Hoffmann S, Meisel A, Attarian S, et al. Efgartigimod demonstrates consistent magnitude of response across subgroups of patients with gMG [abstract no. EPO-054 plus poster]. Eur J Neurol. 2023;30(Suppl. 1):357–8.
- 25. Bril V, Howard JF Jr, Karam C, et al. Effect of efgartigimod on muscle group subdomains in participants with generalized myasthenia gravis: post hoc analyses of the phase 3 pivotal ADAPT study. Eur J Neurol. 2024;31(e16098):1–9.
- 26. Rozsa C, Saccà F, De Bleecker J, et al. Efgartigimod demonstrates consistent improvements in patients with gMG regardless of prior treatment failures [abstract no. EPO-388 plus poster]. Eur J Neurol. 2023;30(Suppl. 1):568.
- Bril V, Vu T, Brauer E, et al. Efgartigimod demonstrates consistent improvements in generalized myasthenia gravis patients of shorter disease duration [abstract no. P1–5.015 plus poster]. Neurology. 2023;100(17 Suppl. 2):4070.
- Pulley M, Pasnoor M, Gelinas D, et al. The effect of obesity in efficacy and safety in the ADAPT trial of efgartigimod for generalized myasthenia gravis [poster no. 194]. In: American Association of Neuromuscular & Electrodiagnostic Medicine (AANEM) Annual Meeting. 2022.
- 29. Saccà F, Barnett C, Vu T, et al. Efgartigimod improved healthrelated quality of life in generalized myasthenia gravis: results from a randomized, double-blind, placebo-controlled, phase 3 study (ADAPT). J Neurol. 2023;270(4):2096–105.
- Dewilde S, Qi CZ, Phillips G, et al. Association between myasthenia gravis-activities of daily living (MG-ADL) and EQ-5D-5L utility values: the additional effect of efgartigimod on utilities. Adv Ther. 2023;40(4):1818–29.
- Katyal N, Halldorsdottir K, Govindarajan R, et al. Safety and outcomes with efgartigimod use for acetylcholine receptor-positive generalized myasthenia gravis in clinical practice. Muscle Nerve. 2023;68(5):762–6.
- 32. Qi C, Narayanaswami P, Wolfe G, et al. Real-world outcomes of patients living with generalized myasthenia gravis initiating efgartigimod treatment in the United States [poster]. In: American Association of Neuromuscular & Electrodiagnostic Medicine (AANEM) Annual Meeting. 2023.
- 33. Singer M, Khella S, Bird S, et al. Single institution experience with efgartigimod in patients with myasthenia gravis: patient selection, dosing schedules, treatment response, and adverse events. Muscle Nerve. 2024;69(1):87–92.
- Moniz Dionísio J, Ambrose P, Burke G, et al. Efgartigimod efficacy and safety in refractory myasthenia gravis: UK's first real-world experience. medRxiv. 2024. https://doi.org/10.1101/ 2024.01.31.24302082.
- 35. Frangiamore R, Rinaldi E, Vanoli F, et al. Efgartigimod in generalized myasthenia gravis: a real-life experience at a national reference center. Eur J Neurol. 2024;e16189:1–10.
- Pane C, Di Stefano V, Cuomo N, et al. A real-life experience with eculizumab and efgartigimod in generalized myasthenia gravis patients. J Neurol. 2024. https://doi.org/10.1007/ s00415-024-12588-7.
- Wang P, Zhang B, Yin J, et al. Prospective cohort study evaluating efficacy and safety of efgartigimod in Chinese generalized myasthenia gravis patients. Front Neurol. 2024;15:1407418.
- Suzuki S, Uzawa A, Nagane Y, et al. Therapeutic responses to efgartigimod for generalized myasthenia gravis in Japan. Neurol Clin Pract. 2024;14(3): e200276.
- 39. Goyal N, Qi C, Stone J, et al. Real-world reduction in oral corticosteroid utilization following efgartigimod initiation in patients living with generalized myasthenia gravis [oral presentation]. In: American Academy of Neurology (AAN) Annual Meeting. 2024.

- 40. Howard Jr JF, Li Y, Vu T, et al. Long-term safety, tolerability, and efficacy of subcutaneous efgartigimod PH20 in participants with generalized myasthenia gravis: interim results of the ADAPT-SC+ study [poster 008]. In: American Academy of Neurology (AAN) Annual Meeting. 2024.
- 41. Howard Jr JF, Vu T, De Bleecker J, et al. Long-term safety, tolerability, and efficacy of subcutaneous efgartigimod PH20 in participants with generalized myasthenia gravis: interim analysis of anti-acetylcholine receptor autoantibody seronegative participants in the ADAPT-SC+ study [poster no. 151]. In: American Association of Neuromuscular & Electrodiagnostic Medicine (AANEM) Annual Meeting. 2023.
- 42. Vu T, Howard Jr J, Korobko D, et al. Subcutaneous efgartigimod PH20 treatment in participants with generalized myasthenia gravis in ADAPT-SC+: interim analyses on quality of life, efficacy, tolerability, and long-term safety [presentation]. In: American Association of Neuromuscular & Electrodiagnostic Medicine (AANEM) Annual Meeting. 2023.
- 43. Podhorna J, Ward C, Lee I, et al. Analysis of serious infections and malignancy risk in myasthenia gravis: a US claims database study [poster 007]. In: American Academy of Neurology (AAN) Annual Meeting. 2024.
- Sanders DB, Wolfe GI, Benatar M, et al. International consensus guidance for management of myasthenia gravis: executive summary. Neurology. 2016;87(4):419–25.
- 45. Hoy SM. Rozanolixizumab: first approval. Drugs. 2023;83(14):1341-7.
- Dhillon S. Eculizumab: a review in generalized myasthenia gravis. Drugs. 2018;78(3):367–76.
- 47. Kang C. Ravulizumab: a review in generalised myasthenia gravis. Drugs. 2023;83(8):717–23.
- 48. Shirley M. Zilucoplan: first approval. Drugs. 2024;84(1):99–104.
- 49. Fenton C, Kang C. Several more myasthenia gravis therapy options now available. Drugs Ther Perspect. 2024;40:350–2.
- 50. Wiendl H, Abicht A, Chan A, et al. Guideline for the management of myasthenic syndromes. Ther Adv Neurol Disord. 2023;16:1–31.
- 51. Gilhus NE, Andersen H, Andersen LK, et al. Generalized myasthenia gravis with acetylcholine receptor antibodies: a guidance for treatment. Eur J Neurol. 2024;e16229:1–13.
- 52. Bril V, Hussain Y, Gwathmey K, et al. Fixed cycle and everyother-week dosing of intravenous efgartigimod for generalized myasthenia gravis: part A of ADAPT NXT [poster 010]. In: American Academy of Neurology (AAN) Annual Meeting. 2024.
- 53. Chen H, Qiu Y, Yin Z, et al. Efficacy and safety of the innovative monoclonal antibodies in adults with generalized

myasthenia gravis: a Bayesian network analysis. Front Immunol. 2023;14(1280226.):1–14.

- 54. Ma Y, Nie X, Zhu G, et al. The efficacy and safety of different targeted drugs for the treatment of generalized myasthenia gravis: a systematic review and Bayesian network meta-analysis. CNS Drugs. 2024;38:93–104.
- Meuth S, Hagenacker T, Scheiner C, et al. Matching-adjusted indirect comparison of ravulizumab/efgartigimod in generalised myasthenia gravis: timepoint challenges [abstract no. EPO-225]. Eur J Neurol. 2023;30(Suppl. 1):467.
- Saccà F, Pane C, Espinosa PE, et al. Efficacy of innovative therapies in myasthenia gravis: a systematic review, meta-analysis and network meta-analysis. Eur J Neurol. 2023;30(12):3854–67.
- 57. Wolfe G, Saccà F, Phillips G, et al. Network meta-analysis of treatment options in generalized myasthenia gravis: impact on healthrelated quality of life outcome [poster]. In: American Association of Neuromuscular & Electrodiagnostic Medicine (AANEM) Annual Meeting. 2023.
- Sacca F, Dewilde S, Nivelle E, et al. Combining clinical trial and real-world data to model the benefit of efgartigimod on productivity losses [abstract no. PS05.02]. J Neuromuscul Dis. 2022;9(Suppl.):S108-9.
- Qi C, Wang J, Yang H, et al. Numbers needed to treat and costs per improved outcome among treatments for myasthenia gravis [abstract no. EE121 plus poster]. Value Health. 2022;25(7):S358.
- 60. Siddiqi Z, Genge A, Qi CZ, et al. Cost-effectiveness analysis of efgartigimod versus chronic intravenous immunoglobulin (IVIg) for treatment of acetylcholine receptor antibody positive (AChR-Ab+) generalized myasthenia gravis (gMG) in Canada [poster P4.003]. In: American Academy of Neurology (AAN) Annual Meeting. 2024.
- 61. Tice JA, Touchette DR, Lien PW, et al. The effectiveness and value of eculizumab and efgartigimod for generalized myasthenia gravis. J Manag Care Spec Pharm. 2022;28(1):119–24.
- National Institute for Health and Care Excellence. Draft guidance consultation: efgartigimod for treating generalised myasthenia gravis. 2023. https://www.nice.org.uk/guidance/GID-TA10986/ documents/draft-guidance-3. Accessed 18 Sep 2024.

Springer Nature or its licensor (e.g. a society or other partner) holds exclusive rights to this article under a publishing agreement with the author(s) or other rightsholder(s); author self-archiving of the accepted manuscript version of this article is solely governed by the terms of such publishing agreement and applicable law.